Author's response to reviews

Title: The course of mental health after miscarriage and induced abortion: a longitudinal, five-year follow-up study.

Authors:

Anne Nordal Broen MD (a.n.broen@medisin.uio.no)
Torbjorn Moum Professor (torbjorn.moum@medisin.uio.no)
Anne Sejersted Bodtker MD (abodtker@online.no)
Oivind Ekeberg MD, Professor (oivind.ekeberg@medisin.uio.no)

Version: 4 Date: 1 December 2005

Author's response to reviews:

In this revised manuscript, I have accepted all your suggested changes. The only change I have made, is in the title: I have written "and" instead of "or". This is in accordance with what you too have written earlier. And the title also includes the word "longitudinal". So the correct title is: "The course of mental health after miscarriage and induced abortion: a longitudinal, five-year follow-up study".